Author’s response to reviews

Title: Short and Long-term Barriers and Facilitators of Skin Self-Examination Among Individuals Diagnosed with Melanoma

Authors:

Adina Coroiu (acoroiu@hsph.harvard.edu)
Chelsea Moran (chelsea.moran1@ucalgary.ca)
Catherine Bergeron (catherine.bergeron2@mail.mcgill.ca)
Martin Drapeau (martin.drapeau@mcgill.ca)
Beatrice Wang (Beatrice.Wang@mcgill.ca)
Abbas Kezouh (abbas.kezouh@ladydavis.ca)
Jochen Ernst (jochen.ernst@medizin.uni-leipzig.de)
Gerald Batist (gbatist@onc.jgh.mcgill.ca)
Annett Körner (annett.korner@mcgill.ca)

Version: 2 Date: 10 Dec 2019

Author’s response to reviews:

1. If the need for consent to participate was waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this in the section 'Ethics Approval and Consent to Participate', including the name of the IRB or a reference to the relevant legislation.
   - Added this information on Page 9, last paragraph.

2. Please include a Keywords section below your abstract listing three to ten keywords representing the main content of the article.
   - Added this information on Page 3- Abstract, after Word count

3. In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/submissions/editorial-policies#standards+of+reporting), could you please ensure your manuscript reporting adheres to STROBE guidelines (http://www.strobe-statement.org/) for reporting observational research. This is so your methodology can be fully evaluated and utilised. Can you please include a completed STROBE checklist as an additional file when submitting your revised manuscript.
   - We refer to the STROBE checklist for reporting in our manuscript (Page 9, Method, Study Design subsection). We have also included a STROBE checklist with this revision.

4. Please ensure that all figures are cited in the manuscript text.
- We replaced the word "Appendix" with "Figure 1" on Page 14 - Results section, first paragraph.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.
- We included a clean copy with this revision.